Author’s response to reviews

Title: THE EFFECT OF LOW LEVEL LASER IRRADIATION ON OXIDATIVE STRESS, MUSCLE DAMAGE AND FUNCTION FOLLOWING NEUROMUSCULAR ELECTRICAL STIMULATION. A DOUBLE BLIND, CROSS-OVER TRIAL

Authors:
Ewa Jówko (ewa.jowko@awf-bp.edu.pl)
Maciej Płaszewski (maciej.plaszewski@awf-bp.edu.pl)
Maciej Cieśliński (maciek.cieslinski@awf-bp.edu.pl)
Tomasz Sacewicz (tomasz.sacewicz@awf-bp.edu.pl)
Igor Cieśliński (igor.cieslinski@awf-bp.edu.pl)
Marta Jarocka (marta.jarocka@awf-bp.edu.pl)

Version: 5 Date: 30 Oct 2019

Author’s response to reviews:

Dear Editors,

Thank you very much for your support and assistance in the processing of our submission. We are very satisfied that our paper is now likely to be published in the BMC Sports Science, Medicine and Rehabilitation journal. We very much appreciate the quality of your article processing standards.

Please find below our replies to your comments. We are submitting the final, clear version of the manuscript and figures.

We have made some necessary amendments to our affiliations, as the name and structure of our Faculty have been changed since October, 2019. The main affiliation to our University has not changed, and we are representing the same institution as in the original submission.

With our best wishes,

Sincerely,

The authors
Editor Comments:

1. Please clarify who EJ is to correspond to in the Authors’ contributions subsection as we have not found an author listed on the title page or in the submission system. Please clarify if EJ is a typo or if there is an author missing from the aforementioned lists.

Reply:

EJ is Ewa Jówko, the first and corresponding author. To keep to the Journal’s style, we have decided to correct both the initials (as appearing in the text) and authors’ names in the title page. Now it reads first name – family name (e.g. Ewa Jówko, not Jówko Ewa, and EJ, not JE). We hope this is now clear and logical.

2. At this stage you may remove the CONSORT checklist.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply to 2 and 3: We will act accordingly.

In Figure 1, we changed “randomization” to “randomisation” to keep with British English spelling throughout the manuscript.

Thank you again for your support.

The authors